Author’s response to reviews

Title: Unusual tumours of the heart: diagnostic and prognostic implications.

Authors:

    Zeena Makhija Dr. (profknow171@yahoo.com)
    Ranjit Deshpande Dr. (ranjit.deshpande@kch.nhs.uk)
    Jatin Desai Dr. (jatin.desai@kch.nhs.uk)

Version: 2 Date: 2 January 2009

Author’s response to reviews:

Dear Reviewer,

The issue of case 2 has been addressed. Co-existence of myxoma with aspergillus infection is unusual and has been elaborated in the discussion.

The issue associated with uncertainties of diagnosing and managing such patients has been included in the discussion.

We agree that our case series does not provide any conclusive evidence in this regard and it calls for a multicenter registry.

Yours sincerely,

Authors.